%\title{YOUR TITLE HERE}
\documentclass[11pt]{article}
\usepackage{fancyhdr, amsmath, amssymb}
\usepackage[total={6.5in,9.5in},centering,includefoot,includehead]{geometry}
%\usepackage{graphicx}
\usepackage[svgnames,table]{xcolor}



% FancyHeadings Commands
\renewcommand{\headrulewidth}{0pt}
\cfoot{\small\textit{ Number Theory}}
\lfoot{\small\textit{Math 351}}
\rfoot{\small\textit{ \thepage}}
\lhead{\tiny }\chead{\tiny }\rhead{\tiny  }

\setlength{\parindent}{0pt}


\newcommand{\vertspace}[1]{\vspace*{#1\baselineskip}\par}
\begin{document}
\pagestyle{fancy}


%%%%%%%%%%%%%%%%%% TITLE  %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{center}
Your Name \\
Homework \#X \\
Month XX, 2013
\end{center}

\textit{I have abided  by the Wheaton College Honor Code in this work. } 
-- \textbf{Your Initials}

\vertspace{2.0}
\underline{Problem Number:}  \textit{Statement of problem}

(I worked with XX, YY, and ZZ on this problem.)
\vertspace{0.75}
Solution to problem 


\vertspace{2.0}
\underline{Problem Number:}  \textit{Statement of problem}

(I worked with XX, YY, and ZZ on this problem.) 
\vertspace{0.75}
Solution to problem 


\vertspace{2.0}
\underline{Problem Number:}  \textit{Statement of problem}

(I worked with XX, YY, and ZZ on this problem.)
\vertspace{0.75}
Solution to problem 


	
\end{document}
